Author's response to reviews

Title: Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors:

Jamie C Brehaut (jbrehaut@ohri.ca)
Alison Lott (alott@ohri.ca)
Dean A Fergusson (daferfusson@ohri.ca)
Kaveh G Shojania (kshojania@ohri.ca)
Jonathan Kimmelman (jonathan.kimmelman@mcgill.ca)
Raphael Saginur (rsaginur@ottawahospital.on.ca)

Version: 2 Date: 2 May 2008

Author's response to reviews:

Dear Dr. Foy,

Re: Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors: Jamie C. Brehaut, Alison Lott, Dean A. Fergusson, Kaveh G. Shojania, Jonathan Kimmelman, Raphael Saginur

Thank you for your response regarding the acceptance of the above study protocol for publication in Implementation Science. As per your request, we have addressed two issues regarding the length of the paper. The abstract is now 294 words, so it falls within the word count restriction of the journal. We also followed your discretionary suggestion, and reduced the length of the text; the manuscript now consists of 5934 words.

Please let me know if you require any additional information.

Sincerely,

Jamie C. Brehaut, PhD